Author's response to reviews

Title: Signs and symptoms of temporomandibular joint disorders related to the degree of mouth opening and hearing loss.

Authors:

    Panagiotis Kitsoulis (pkitsoulis@hotmail.com)  
    Aikaterini Marini (katerinaki88@yahoo.gr)  
    Kalliopi Iliou (kallina2@hotmail.com)  
    Vasiliki Galani (vgalani@cc.uoi.gr)  
    Aristides Zimpis (azimpis@hotmail.com)  
    Konstantinos Natsis (konatsis@hotmail.gr)  
    Georgios Paraskevas (g_paraskevas@hotmail.com)

Version: 5  Date: 14 May 2011

Author's response to reviews: see over
Dear Sirs,

Thank you for the final comments and the approval of our manuscript. Here we submit the revised form of it according to your suggestions. More precisely, we made the following changes:

1. We changed “Aims” section of the ABSTRACT into “Background” as you suggested.
2. We now declare that we have no competing interests.
3. We added an “Authors’ contributions” section.
4. All references are in numerical order.
5. All the tables are now incorporated in the main manuscript.
6. On the title page we added the author’s email addresses.
7. We removed the keywords from the abstract section.
8. We renamed Introduction as Background.
9. All the missing authors’ names are now added in the references.
10. We deleted the reference of table 11 from the 5th page.
11. We corrected all the typographic errors throughout the manuscript.
12. Instead of Mr. Natsis, we now include in the authors of the manuscript the Professor and Chairman of our Laboratory, Mr. Kanavaros.

Thank you very much for the cooperation and we remain at your disposal for any further changes or corrections required.

Yours Sincerely,

Prof. P. Kitsoulis